Author's response to reviews

Title: CD133 antigen expression in ovarian cancer

Authors:

Gabriella Ferrandina (gabriella.ferrandina@libero.it)
Enrica Martinelli (enrica.martinelli@rm.unicatt.it)
Marco Petrillo (afpetrillo@libero.it)
Maria Grazia Prisco (mariagrazia.prisco@gmail.com)
Gianfranco Zannoni (gfzannoni@rm.unicatt.it)
Stefano Sioletic (sioletics@hotmail.com)
Giovanni Scambia (giovanni.scambia@rm.unicatt.it)

Version: 3 Date: 21 June 2009

Author's response to reviews:

Rome, June 22, 2009

Dr. Melissa Norton, M.D.
EDITOR IN CHIEF
BMC Cancer

Dear Dr. Norton,

Please find enclosed the electronic version of the manuscript “CD133 antigen expression in ovarian cancer patients” by Ferrandina G. et al., which was revised according to the Referee’s suggestions.

We hope it is now suitable for publication in BMC Cancer.

Yours sincerely,

Gabriella Ferrandina, MD

Reply to Referee 1

1. According to the Referee's suggestion, the manuscript title was changed as follow: “CD133 antigen expression in ovarian cancer”.